\documentstyle[12pt]{article}
\begin{document}
\begin{center}
{\bf NOTE:} office hours instructor: MC6065 x4597 Tues. 2-3PM, Thurs. 1-2PM.\\
~~~~~~~~~~office hours TA Eddie Cheng: MC5165 x2159 Wed. 2-3PM.\\

{\large\bf C\&O 367 \\ 
Assignment MISCELLANEOUS 
}
\end{center}
\begin{flushleft}
{\large  Due on Thursday, Jan. 30,~~~~   Instructor H. Wolkowicz
}
\end{flushleft}

{\bf NOTE:} Several exercises are marked: {\bf Do not hand in.}
\begin{enumerate}

\item
({\bf Do not hand in.}) 
Consider the function
\[f(x,y) = x^2 + y^2 (1-x)^3. \]
Show that this function has a local minimum at the origin and that this
point is the unique stationary point but it is not a global minimum.
\end{enumerate}
\end{document}
